Author's response to reviews

Title: LEARN 2 MOVE 0-2 years: effects of a new intervention program in infants at very high risk for cerebral palsy; a randomized controlled trial.

Authors:

Tjitske Hielkema (t.hielkema@developmentalneurology.com)
Elisa G Hamer (e.g.hamer@student.rug.nl)
Heleen A Reinders-Messelink (h.a.reinders@rev.umcg.nl)
Carel GB Maathuis (c.g.b.maathuis@rev.umcg.nl)
Arend F Bos (a.f.bos@bkk.umcg.nl)
Tineke Dirks (j.f.dirks@developmentalneurology.com)
Lily van Doormaal (e.f.van.doormaal@bkk.umcg.nl)
Johannes MA Verheijden (jverheijden@bosk.nl)
Carla Vlaskamp (c.vlaskamp@rug.nl)
Eline Lindeman (e.lindeman@umcutrecht.nl)
Mijna Hadders-Algra (m.hadders-algra@med.umcg.nl)

Version: 4 Date: 28 October 2010

Author's response to reviews: see over
Dear Editor,

Thank you for your remarks regarding our submitted study protocol ‘LEARN 2 MOVE 0-2 years: effects of a new intervention program in infants at very high risk for cerebral palsy; a randomized controlled trial’ (MS: 1974856282401708), part of the Dutch research program LEARN 2 MOVE.

Conform your suggestions, we made the following changes in our manuscript:

- We changed ‘submitted’ into ‘accepted for publication’ in references 3, 4 and 5 (the other LEARN 2 MOVE study protocols).
- We changed reference 45.
- We changed reference 31 into reference 30 in the text.
- The abbreviations are now explained in a paragraph instead of in a list.
- We removed the blank page 18.
- We removed the vertical lines from the table.

Yours sincerely,

Prof. Dr. Mijna Hadders-Algra